Tutorial for week 12 (Mar 27, 2002)

March 21, 2002

The following questions will be considered during the tutorial sessions on Wednesday...

Q1: (Research) Make up a simple CGI form, similar to the one given on page 71, which uses the POST method for reading the data, and prints out the three fields.

Q2: (Research) Make up a simple PHP processed form, similar to the one given on page 71, and which prints out the three fields.

Q3: Find an example of a site which is using PHP/MySQL with a large database. Give the URL, and a brief note on the site (size of database, type of user interface...).

Q4: Examine the Lissajous.java code. What is the function of the diffx variable?